## Point-by-point responses to reviewers' comment

Sept. 8, 2016

Dear Dr. Ze-Mao Gong,

Please find attached the edited manuscript in Word format (file name: 29207\_revised.doc).

Title: Mesenchymal Stromal Cell-based therapy: regulatory and translational aspects in gastroenterology

Authors: Giovanni Dothel, Emanuel Raschi, Roberto Rimondini and Fabrizio De Ponti

Name of Journal: World Journal of Gastroenterology

ESPS Manuscript NO: 29207

We have revised the manuscript according to the reviewers' suggestions. A point-by-point outline of responses is provided below.

## Reviewer # 02848336.

This is a very well written, comprehensive review paper concerning a problem which is very important and up-to-date. In my opinion the manuscript deserves publication in WJGNET in the present form.

Classification: Grade B (Very good)

Language evaluation: Grade A: priority publishing

Conclusion: Accept

A. We thank the reviewer for these comments.

## Reviewer # 02446114.

An excellent review paper for the readers of the journal, World Journal of Gastroenterology.

Classification: Grade B (Very good)

Language evaluation: Grade B: minor language polishing

Conclusion: High priority for publication

## A. We appreciate reviewer's comments. References and typesetting were corrected.

Thank you again for considering our manuscript in the World Journal of Gastroenterology.

Sincerely yours,



Fabrizio De Ponti, MD, PhD

Pharmacology Unit - Department of Medical and Surgical Sciences - University of Bologna

Via Irnerio, 48

I-40126 Bologna BO - Italy

Tel.: +39-051-2091805 - Fax: +39-051-2091780

e-mail: fabrizio.deponti@unibo.it